Author's response to reviews

Title: Randomised trials relevant to mental health conducted in low and middle-income countries: survey of studies published in 1991, 1995 and 2000 and assessment of their relevance

Authors:

Rebecca J Syed Sheriff (rebecca_syd@hotmail.com)
Mahesh Jayaram (maheshbj@gmail.com)
Prathap Tharyan (prathap@cmcvelllore.ac.in)
Lelia Duley (lelia.duley@ndm.ox.ac.uk)
Clive E Adams (ceadams@cochrane-sz.org)

Version: 4 Date: 22 September 2006

Author's response to reviews:

To Whom It May Concern,

I have now made the formatting change requested,

thanks